Author’s response to reviews

Title: Talking Health: trusted health messengers and effective ways of delivering health messages for rural Mothers in Southwest Ethiopia

Authors:

shifera Asfaw (shiferaa@yahoo.com)
Morankar Sudhakar (morankarsn@yahoo.com)
Muluemebet Abera (mulu_bera.ts2009@yahoo.com)
Abebe Mamo Gebretsadik (abema24@yahoo.com)
Lakew Abebe Gebretsadik (lakewgebretsadika9@gmail.com)
Nicole Bergen (nicolejbergen@gmail.com)
Manisha Kulkarni (manisha.kulkarni@uottawa.ca)
Ronald Labonté (rlabonte@uottawa.ca)

Version: 2 Date: 17 Jan 2019

Author’s response to reviews:

Both reviewers accept the manuscript for publication.

BMC archive of public health
AOPH-D-18-00133

Talking Health: trusted health messengers and effective ways of delivering health messages for rural Mothers in Southwest Ethiopia

Shifera Asfaw1*, Sudhakar Morankar1, Muluemebet Abera1, Abebe Mamo1, Lakew Abebe1, Nicole Bergen2, Manisha Kulkarni2, Ronald Labonté2

Response to Editor

First we want to express our heart full acknowledgement for the editor. The forwarded suggestions are highly valuable to improving the manuscript.
We agreed and revised the manuscript accordingly.

Response to the reviewers

All authors appreciate your great impute on the current manuscript.

Thank you!